

This is a repository copy of Insights from ADPedKD, ERKReg and RaDaR registries provide a multi-national perspective on the presentation of childhood autosomal dominant polycystic kidney disease in high- and middle-income countries.

White Rose Research Online URL for this paper: <u>https://eprints.whiterose.ac.uk/id/eprint/227806/</u>

Version: Accepted Version

Article:

Gimpel, C. orcid.org/0000-0003-1296-9081, Fieuws, S., Hofstetter, J. et al. (146 more authors) (2025) Insights from ADPedKD, ERKReg and RaDaR registries provide a multinational perspective on the presentation of childhood autosomal dominant polycystic kidney disease in high- and middle-income countries. Kidney International. ISSN 0085-2538

https://doi.org/10.1016/j.kint.2025.02.026

© 2025 The Authors. Except as otherwise noted, this author-accepted version of a journal article published in Kidney International is made available via the University of Sheffield Research Publications and Copyright Policy under the terms of the Creative Commons Attribution 4.0 International License (CC-BY 4.0), which permits unrestricted use, distribution and reproduction in any medium, provided the original work is properly cited. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/

Reuse

This article is distributed under the terms of the Creative Commons Attribution (CC BY) licence. This licence allows you to distribute, remix, tweak, and build upon the work, even commercially, as long as you credit the authors for the original work. More information and the full terms of the licence here: https://creativecommons.org/licenses/

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



Graphical Abstract

A multi-national perspective on the presentation of childhood ADPKD in high- and middle-income countries: insights from the ADPedKD, ERKReg and RaDaR registries





Gimpel, 2024

On behalf of the ADPedKD, ERKReg & RaDaR Consortia

adpedkd.org erknet.org X@ADPedKD @EURefNetwork

ukkidney.org etwork **CONCLUSION** High rate of children diagnosed by active screening. Prenatal diagnoses rise after 2000. Greater geographical than temporal variations show influence of local health care systems. A multi-national perspective on the presentation of childhood autosomal dominant polycystic kidney disease in high- and middle-income countries: insights from the ADPedKD, ERKReg and RaDaR registries

Running title: Global presentation of childhood ADPKD

Type of manuscript: Original Article, Clinical Investigation

Corresponding author:

Djalila Mekahli, MD, PhD Department of Pediatric Nephrology University Hospitals Leuven Herestraat 49 B-3000 Leuven, Belgium Email: <u>djalila.mekahli@uzleuven.be</u> Tel: +32 16 34 38 22

Fax: +32 16 34 38 42

Word count abstract: 250 words, 1699 characters Word count article (without references, tables and figures): 4544

Authors

Charlotte Gimpel MD PhD¹, Steffen Fieuws PhD^{2,3}, Jonas Hofstetter BSc MSc¹, David Pitcher MD^{4,5}, Lotte Vanmeerbeek^{2,6}, Stefanie Haeberle PhD¹, Angélique Dachy MD^{2,6,7}, Laura Massella MD⁸, Tomas Seeman MD, PhD^{9,10}, Bruno Ranchin MD¹¹, Lise Allard MD¹², Justine Bacchetta MD, PhD¹¹, Umut S. Bayrakci MD, PhD¹³, Francesca Becherucci MD, PhD^{14,15}, Victor Perez-Beltran MD¹⁶, Martine Besouw MD, PhD¹⁷, Hanna Bialkevich MD, PhD¹⁸, Olivia Boyer MD, PhD^{19,20}, Nur Canpolat MD, PhD²¹, Dominique Chauveau MD, PhD²², Neslihan Cicek MD²³, Peter J Conlon MD²⁴, PhD, Olivier Devuyst MD, PhD²⁵, Claire Dossier MD²⁶, Marc Fila MD, PhD²⁷, Hana Flögelová MD, PhD²⁸, Astrid Godron-Dubrasquet MD¹², Ibrahim Gokce MD²³, Elsa Gonzalez Nguyen-Tang MD²⁹, Juan David González-Rodríguez MD³⁰, Anne Guffens MD³¹, Giuseppe Grandaliano MD, PhD³², Laurence Heidet MD, PhD^{19,20}, Augustina Jankauskiene MD, PhD³³, Tanja Kersnik Levart MD, PhD³⁹, Bertrand Knebelmann MD, PhD^{34,35}, Jens Christian König MD³⁶, Claudio La Scola MD⁴⁰, Valentina Fanny Leone MD³⁷, Valérie Leroy MD³⁸, Mieczyslaw Litwin MD, PhD⁴⁴, Laura Lucchetti MD⁸, Adrian C. Lungu MD, PhD⁴¹, Pierluigi Marzuillo MD, PhD⁴², Antonio Mastrangelo MD⁴³, Monika Miklaszewska MD⁴⁵, Giovanni Montini MD, PhD^{46,47} François Nobili MD⁴⁸, Lukasz Obrycki MD⁴⁴, Svetlana Papizh MD⁴⁹, Aleksandra Paripović MD⁵⁰, Dušan Paripović MD, PhD⁵¹, Licia Peruzzi MD, PhD⁵², Ann Raes MD, PhD^{53,54}, Seha Saygili MD²¹, Brankica Spasojević MD, PhD⁵¹, Thomas Simon MD⁵⁵, Maria Szczepańska MD, PhD⁵⁶, Francesco Trepiccione MD, PhD⁵⁷, Nataša Marčun Varda MD, PhD⁵⁸, Rik Westland MD, PhD⁵⁹, Selcuk Yüksel MD, PhD⁶⁰, Iga Zaluska- Lesniewska MD⁶¹, Julie Tenebaum MD²⁷, Reem Mustafa MD, PhD^{62,63}, Andrew J. Mallett MD, PhD^{64,65,66,67}, Lisa M. Guay-Woodford MD⁶⁸, Daniel P. Gale MD, PhD^{4,5}, Detlef Böckenhauer MD, PhD^{2,4,6}, Max C. Liebau MD, PhD⁶⁹, Franz Schaefer MD, PhD¹, Djalila Mekahli MD, PhD^{2,6}, The RaDaR disease group, the ERKReg and the ADPedKD collaborators.

Affiliations

¹Division of Pediatric Nephrology, Center for Pediatrics and Adolescent Medicine, Heidelberg, Germany

²PKD Research Group, Laboratory of Ion Channel Research, Dept of Cellular and Molecular Medicine, KU Leuven, Leuven, Belgium

³Biostatistics and Statistical Bioinformatics Center, KU Leuven, Belgium

⁴Department of Renal Medicine, University College London, UK

⁵National Registry of Rare Kidney Diseases, Bristol, UK

⁶Dept of Pediatric Nephrology, University Hospital Leuven, Belgium

⁷Department of Pediatrics, ULiège Academic Hospital, Liège, Belgium

⁸Division of Nephrology, Bambino Gesù Children's Hospital, IRCCS, Rome, Italy

⁹Department of Pediatrics, Charles University Prague, 2nd Medical Faculty, Prague, Czech Republic

¹⁰Department of Pediatrics, University of Ostrava, Medical Faculty, Ostrava, Czech Republic
¹¹Pediatric Nephrology Unit, Hôpital Femme Mère Enfant, Hospices Civils de Lyon,
INSERM1033 Research Unit, Université de Lyon, European Rare Kidney Disease Reference
Network Center, Lyon, France.

¹²Reference center for Rare Kidney Diseases (SoRare), ERKNet, Department of Pediatrics, Bordeaux University Hospital, Bordeaux, France

¹³Department of Paediatric Nephrology, Ankara City Hospital, Üniversiteler Mahallesi Bilkent Caddesi, Çankaya/Ankara, Turkey

¹⁴Nephrology and Dialysis Unit, Meyer Children's Hospital IRCCS, Florence, Italy

¹⁵Department of Biomedical Experimental and Clinical Sciences, University of Florence, Italy ¹⁶Paediatric Nephrology Department, University Hospital Vall d'Hebron, Barcelona, Spain

¹⁷Department of Pediatric Nephrology, University Medical Center Groningen, University of Groningen, Groningen, The Netherlands

¹⁸Belarusian State Medical University, Minsk, Republic of Belarus

¹⁹LH, Service de Néphrologie Pédiatrique, Centre de Référence MARHEA, Hôpital Universitaire Necker-Enfants Malades, Assistance publique, Hôpitaux de Paris (AP-HP), Paris, France

²⁰Laboratoire des Maladies Rénales Héréditaires, Inserm UMR 1163, Institut Imagine, Université de Paris-Cité, Paris, France.

²¹Department of Paediatric Nephrology, Istanbul University-Cerrahpasa, Cerrahpasa Faculty of Medicine, Istanbul, Turkey.

²²University Hospital, Toulouse, France

²³Marmara University, Medical Faculty, Division of Pediatric Nephrology, Istanbul, Turkey

²⁴Beaumont Hospital, Dublin, Ireland

²⁵Université catholique de Louvain, Belgium

²⁶Pediatric Nephrology Department, Robert-Debre Hospital, APHP, Paris, France

²⁷Pediatric Nephrology, Montpellier University Hospital, SORARE Reference Center, Montpellier, France

²⁸Department of Pediatrics, Faculty of Medicine and Dentistry, Palacky University Olomouc and University Hospital Olomouc, Czech Republic

²⁹Pediatric Nephrology Unit, Department of Pediatrics, University of Geneva Hospital, Geneva, Switzerland

³⁰Pediatric Nephrology Department, Hospital Universitario Santa Lucía, Cartagena, Spain
 ³¹Service de Pédiatrie, CHC MontLégia, Liège, Belgium

³²Dept of Translational Medicine and Surgery, Università Cattolica del Sacro Cuore, Nephrology Unit, Fondazione Policlinico Universitario A. Gemelli IRCCS, Rome, Italy

³³Pediatric Center, Institute of Clinical Medicine, Vilnius university, Vilnius, Lithuania

³⁴Université Paris Cité, INSERM U1151, Institut Necker Enfants Malades, F-75015 Paris, France

³⁵Service de Nephrologie Adultes, AP-HP, Hôpital Necker, Paris, France

³⁶ Department of General Pediatrics, University Children's Hospital Münster; Münster, Germany

³⁷Unit of Nephrology, Azienda Socio Sanitaria Territoriale (ASST) Papa Giovanni XXIII, Bergamo, Italy

³⁸Pediatric nephrology Unit, CHU La Réunion, Saint Denis, France

³⁹University Medical Centre Ljubljana, University Children's Hospital, Nephrology Department, Ljubljana, Slovenia

⁴⁰Pediatric Nephrology, Pediatric unit, IRCCS Azienda Ospedaliero-Universitaria di Bologna
 ⁴¹Pediatric Neohrology, Fundeni Clinical Institute, Bucharest, Romania

⁴²Department of Woman, Child and of General and Specialized Surgery, Università degli Studi della Campania "Luigi Vanvitelli", Naples, Italy

⁴³Pediatric Nephrology, Dialysis and Transplant Unit, Fondazione IRCCS, Ca' Granda Ospedale Maggiore Policlinico, Milano, Italy

⁴⁴Department of Nephrology, Kidney Transplantation and Arterial Hypertension, The Children's Memorial Health Institute, Warsaw, Poland

⁴⁵Department of Paediatric Nephrology and Hypertension, Jagiellonian University, Medical College of Cracow, Cracow, Poland

⁴⁶Pediatric Nephrology, Dialysis and Transplant Unit, Fondazione IRCCS, Ca' Granda Ospedale Maggiore Policlinico di Milano, Milan, Italy

⁴⁷Giuliana and Bernardo Caprotti Chair of Pediatrics, Department of Clinical Sciences and Community Health, University of Milano, Milan, Italy

⁴⁸Service de Pédiatrie 2, CHU Besançon, Besançon, France

⁴⁹Veltishev Research Clinical Institute for Pediatrics & Children Surgery, Pirogov Russian National Research Medical University, Moscow, Russia

⁵⁰Institute for Mother and Child Health Care, Belgrade, Serbia

⁵¹University of Belgrade Faculty of Medicine, University Children's Hospital, Belgrade, Serbia ⁵²Peditaric Nephrology Unit, Regina Margherita Children's Hospital, Turin, Italy

⁵³Department of Pediatric Nephrology, Ghent University Hospital, Ghent, Belgium,

⁵⁴Department of Internal Medicine and Pediatrics, Ghent University, Ghent, Belgium

⁵⁵Center de Reference de Maladie Rares du Sud-Ouest SORAR, CHU Toulouse, France

⁵⁶Department of Pediatrics, FMS in Zabrze, Medical University of Silesia, Katowice, Poland

⁵⁷Department of Medical Translational Sciences, University of Campania, Naples, Italy

⁵⁸Department of Paediatrics, University Medical Centre Maribor, Maribor, Slovenia

⁵⁹Department of Pediatric Nephrology, Amsterdam UMC-Emma Children's Hospital, University of Amsterdam, Amsterdam, The Netherlands.

⁶⁰ Department of Pediatric Nephrology and Pediatric Rheumatology, School of Medicine, Canakkale Onsekiz Mart University, Canakkale, Turkey

⁶¹Department of Pediatric Nephrology and Hypertension, Medical University of Gdańsk, Gdańsk, Poland

⁶²Department of Clinical Epidemiology and Biostatistics, McMaster University, Hamilton, Ontario, Canada

⁶³Division of Nephrology and Hypertension and the Kidney Institute, University of Kansas Medical Center, Kansas City, USA

⁶⁴ Department of Renal Medicine, Townsville University Hospital, Townsville, Australia

⁶⁵ Institute for Molecular Bioscience, The University of Queensland, Brisbane, Australia

⁶⁶The KidGen Collaborative, Australian Genomics, Melbourne, Australia

⁶⁷College of Medicine and Dentistry, James Cook University, Townsville, Australia

⁶⁸Center for Translational Research, Children's National Research Institute, Washington, DC, USA

⁶⁹Department of Pediatrics and Center for Molecular Medicine Cologne, University of Cologne, Faculty of Medicine and University Hospital Cologne, Cologne, Germany

ABSTRACT

Data on the presentation of Autosomal Dominant Polycystic Kidney Disease (ADPKD) in children has been based on small/regional cohorts and practices regarding both asymptomatic screening in minors and genetic testing differ greatly between countries.

We analyzed over 2100 children and adolescents with ADPKD from 32 countries in 6 WHO world regions: 1060 children from the multi-national ADPedKD registry were compared to pediatric patients from the UK (RaDaR, n=269) and the European Rare Kidney Disease Registry (ERKReg, n=825).

Asymptomatic family screening was a common mode of presentation (48% in ADPedKD, 62% in ERKReg) with broad international variability (19%-75%), but fairly stable temporal trends in both registries and no correlation to genetic testing.

The national rates of genetic testing were very different and correlated significantly with healthcare expenditure (odds ratio 1.030 per 100USD/capita/year, P = 0.002 in the ERKReg cohort), but had little variation over time.

Diagnosis due to prenatal abnormalities was more common than anticipated at 14% (8%-19%); it increased steadily from 2000 onward in both registries.

In summary, a high proportion of children were diagnosed with ADPKD by active screening, underlining that families affected by ADPKD have a high need for counselling on the complex issues around presymptomatic diagnosis. Regional variations in rate of genetic testing appeared to be driven by economic factors, while the rate of active screening was very different between countries and may reflect the influence of different of cultural, legal and ethical frameworks on families and clinicians in different healthcare systems more than economic factors alone.

Keywords: Autosomal dominant polycystic kidney disease, children, adolescents, registry, phenotype, genetic testing, demographics

TRANSLATIONAL STATEMENT

This current real-world map of the way that infants, children and adolescents with ADPKD are diagnosed demonstrates the large influence of regional practices on the complex ethical questions around diagnosing ADPKD in asymptomatic minors. This highlights the need for progress in specialized multidisciplinary counselling of affected families. Further longitudinal analyses should aim to characterize the benefits and harms of early asymptomatic diagnosis. In view of recent advances in delaying disease progression in high-risk individuals, the registries serve as a base for further studies to identify and validate primary endpoints, prognostic factors and predictive biomarkers for children with ADPKD in order to tailor therapy to individual risk.

LAY SUMMARY

Autosomal Dominant Polycystic Kidney Disease (ADPKD) is an inherited disease, characterized by growth of cysts in the kidneys and usually does not progress to kidney failure until adulthood. Less is known about childhood ADPKD and actively looking for the disease before symptoms (screening) is controversial. Ultrasound screening or genetic testing (where accessible) should only be done in minors after careful consideration and discussion with the family about the benefits and harms of presymptomatic screening. We describe how >2100 children from three different registries (ADPedKD, ERKReg, RaDaR) were first diagnosed.

Genetic testing was performed in 52% in ADPedKD and 41% in ERKReg, and differences between countries were related to healthcare spendings.

About half of the children were diagnosed by active screening, but the large international variations were not related to rate of genetic testing.

Diagnoses before birth increased in both ADPedKD and ERKReg from 2000 onwards, probably due to advances in prenatal diagnostics.

In summary, countries with higher healthcare spendings were more likely to perform genetic testing, but the large regional differences in asymptomatic screening also appear to have other influences such as local health-care systems, laws, cultural attitudes or beliefs of the clinicians and families affected by ADPKD.

ABBREVIATIONS

ADPKD	Autosomal dominant polycystic kidney disease
KidGen	Kidneys, Genes and Generations Collaborative (Australia)
NIH-HRFD	NIH-funded Hepato-Renal Fibrocystic Disease Core Resource (North America)
RaDaR	National Registry of Rare Kidney Diseases (United Kingdom)
GDP	Gross domestic product
UN	United nations
USD	US dollars
SD	Standard deviation
IQR	Interquartile range
WHO	World health organization

INTRODUCTION

Autosomal dominant polycystic kidney disease (ADPKD) is the most common inherited kidney disease and the fourth most common cause for kidney replacement therapy in adults worldwide. ADPKD represents a major socio-economic burden for patients, families, communities, and health care systems^{1–4}. Although it is recognized that (mainly microscopic) cyst formation already starts *in utero*⁵, and hypertension is a common feature in young adults⁶, ADPKD often causes few symptoms until the 3rd-4th decade associated with a critical threshold for progression of kidney damage. There is a large phenotypic spectrum of childhood ADPKD which includes incidental sonographic findings of kidney cysts in many cases, but also severe neonatal presentations (resembling autosomal recessive PKD⁷) as well as mild symptomatic courses, with arterial hypertension and/or symptoms (such as macrohematuria, abdominal pain or enuresis) ^{8–11}. Such a broad clinical spectrum combined with inter- and intra-familiar variability of disease progression is challenging for the collection of epidemiological and demographic data^{12,13}. Early nephroprotective strategies are a promising goal for preventing kidney damage¹⁴, so there is an increasing need to understand and characterize disease manifestations of early ADPKD to select those at-risk of rapid progression^{15,16}. Therefore, characterizing the initial mode of presentation of a large and diverse population is an important first step to map current practices and to understand the impact of different local policies.

Despite the fact that several recent initiatives have sought to standardize the management of children with ADPKD^{17–21}, these recommendations need to be integrated into different health care systems in the context of local policies, legal frameworks, and financial constraints as well as diverse cultures, beliefs and apprehensions of families and healthcare staff. We therefore hypothesized that there are still large discrepancies regarding the management and diagnosis of children with ADPKD around the globe, e.g. with respect to proportion of children diagnosed by active screening or rates of genetic testing. Our aim was to analyze the baseline data of the largest pediatric ADPKD registry (ADPedKD) regarding age at diagnosis, mode of presentation and clinical characteristics of children at presentation and

evaluated differences between era of diagnosis, year of birth as well as countries and their socioeconomic characteristics. To extend the geographical spectrum, these data were compared to the UK-based pediatric ADPKD patients collected in the RaDaR registry²² and in the European Rare Kidney Disease Registry (ERKReg)²³.

METHODS

Study population and data collection

We analyzed baseline characteristics of patients included in the multi-national ADPedKD registry (https://www.adpedkd.org)²⁴ until June 2021. The ADPedKD registry collects data on children diagnosed with ADPKD up to the age of 19 years in Europe, Africa and Asia in a common database. Details about the structure, data collection and management of ADPedKD have been previously described²⁴. Diagnosis of ADPKD was based on family history and imaging findings and/or molecular genetic testing. After informed consent, deidentified data were stored in secure online databases. Later the United Kingdom, North and South America, and Australia joined the initiative with their own respective regional databases, namely the UK's National Registry of Rare Kidney Diseases (RaDaR)²⁵, NIH-funded Hepato-Renal Fibrocystic Disease (HRFD) Core Resource²⁶ and KidGen Collaborative (KidGen)²⁷. For the present analysis, the NIH-HRFD and KidGen datasets were merged with the original ADPedKD registry, while patients from the RaDaR database are used as a pediatric control group, mainly because patient level-data was not available for genetic testing, nor mode of diagnosis for RaDaR. There were no duplicate patients. The second control group was extracted from ERKReg, the investigator-led registry of the European Reference Network for rare kidney diseases (ERKNet) funded by the European Union to which both adult and pediatric nephrology centers contribute ²³. Eligible patients had a clinical and/or genetic diagnosis of ADPKD before the age of 19 years and were not also included in the ADPedKD registry. Data were extracted in October 2023.

Four modes of presentation were defined: (1) at-risk children from families with known ADPKD identified by active screening (family screening), (2) incidental finding of kidney cysts

on abdominal ultrasound in a child (postnatal incidental finding), (3) prenatal diagnosis of ADPKD by ultrasound screening in pregnancy (prenatal diagnosis) and (4) children who presented with signs and/or symptoms related to ADPKD such as gross hematuria, pain, hypertension, or proteinuria (clinical signs/symptoms). In ADPedKD the mode of presentation was classified by the local investigator, even though clinical symptoms were not always unequivocally attributable to ADPKD and some patients with later symptomatic or incidental diagnoses had records of prenatal abnormalities. In ERKReg mode of presentation was defined according to the answers regarding previous clinical symptoms, known family history, method by which diagnosis was made, and prenatal ultrasound findings. For patients with a prenatal diagnosis, the age at diagnosis was expressed as a negative value corresponding to the time before birth at which anomalies were first noted.

For analysis of local geographical effects, countries were grouped according to world health organization (WHO) regions²⁸ and Europe further subdivided according to the UN geoscheme²⁹. Country-specific healthcare expenditures, expressed both as US dollar (USD) per capita or as a percentage of gross domestic product (GDP), were taken from the data repository from the WHO for 2015^{30,31}.

Statistical analysis

A validation plan with automated quality checks for coherence of the submitted data was performed in both cohorts. Direct contact between responsible statisticians ensured consistent and comparable data analyses across cohorts. Descriptive statistics are given as relative and absolute frequencies of all non-missing cases for binary/categorical variables and of mean, standard deviation (SD), median and interquartile range (IQR) for the continuous variables. Unknown data were treated as missing.

To obtain country- and region-specific estimates for the probability that a genetic test was performed and for the probabilities of each mode of presentation, generalized linear mixed models (i.e., binary logistic regression models with a random country/region effect) were used. From these models empirical Bayes estimates and 95% confidence intervals for each country/

region were derived³².Note that the smaller the number of subjects in a region/country, the stronger the country-specific Bayes estimate will shift towards the overall mean of all regions/ all countries.

To assess the determinants of age at diagnosis univariable linear mixed models were used for era of diagnosis, genetic test performed, mode of presentation, positive family history and genotype, adding a random effect of country to handle the correlation between patients of the same country. A linear mixed model only with country as random effect was used to evaluate differences between countries. Consequently, several factors were combined without model reduction into a multivariable linear mixed model based on *a priori* variable selection. Chi²-tests were used to explore univariable relations with genetic test performed and modes of presentation. All analyses were performed using SAS software, version 9.4 of the SAS System for Windows by senior statisticians (SF and JH) of the respective registries.

Ethical Statement

The study was approved by the ethics committee of University Hospitals Leuven (S59638) as the leading center of ADPedKD and confirmed by the local institutional review boards of all participating centers or through the protocols for the participating regional databases. ERKReg, RaDaR, NIH and KidGen registries and their informed consent forms were approved by the local institutional review boards as required by local and national regulations. The study was conducted according to the 2013 Declaration of Helsinki, the guidelines of Good Clinical Practice, and all applicable regulatory requirements, including the European Union General Data Protection Regulation.

RESULTS

We analyzed 2154 children and adolescents with ADPKD from 32 countries on four continents. From the ADPedKD registry, data of 1060 children with ADPKD from 25 countries and 63 centers were analyzed, including 17 (1.6%) from KidGen and 16 (1.5%) from HRFD. Of these 547 (52%) were females. The RaDaR cohort comprised 269 patients (57%) females)

and the ERKReg cohort 825 patients (51% females), of which 390 (47%) came from adult centers. Patients originated from 5 of the 6 WHO world regions, with 24 countries being rated as "high income" and 8 as "middle income" with a range of yearly healthcare expenditure from 58 to 9,383 USD per captia. Patient numbers per region and weighted mean healthcare expenditure of each region is given in Table 1. The number of patients per country is given in Supplementary Table S1.

The data regarding demographics, mode of presentation and genetic findings for each registry are presented in Table 2. ADPKD was genetically confirmed in the child themself in 50% of the ADPedKD cohort (525 of 1060, see Table 2). The rate of genetic confirmation in the child was lower in the RaDaR and ERKReg cohorts (30% and 33%), mainly due to a higher percentage of children with missing data on genetic results (31% and 18% vs 4% in ADPedKD), but also because of a lower rate of genetic testing (43% and 41% vs 52% respectively). The ratio of *PKD1* to *PKD2* causative variants was similar to published adult cohorts with 10.6 to 1.0 in the ADPedKD group and similar in RaDaR (12.5 to 1.0) and ERKReg cohorts (9.0 to 1.0). There was a known positive family history for ADPKD in 79% of ADPedKD children (825 of 1044), 72% of ERKReg children (595 of 825) and 97% of children (231 of 238) of RaDaR children.

Mode and age of presentation

In the group of 950 children with data available on mode of presentation from the ADPedKD cohort, nearly half were diagnosed by family screening (455/950, 48%), while the second most common mode was postnatal incidental finding (293/950, 31%), followed by prenatal diagnosis in 129/950 (14%) and clinical signs or symptoms in 73/950 (8%, see Table 3). There was no significant difference in mode of presentation between boys and girls. About half (n=429/764 (56.2%)) of patients with a positive family history were diagnosed via family screening, while 12% (92/764) were diagnosed prenatally, 7% (52 of 764) with clinical symptoms and 25% (191/764) with postnatal incidental findings. The rate of genetic testing was similar in the family screening group (238/455 (52%)) compared to the symptomatic diagnosis group (41/73 (56%)) and the incidental finding group (142/293 (48%)) (P = 0.057).

By comparison, in ERKReg of the 595 patients with a positive family history 370 (86%) were diagnosed by family screening, 3 (0,7%) prenatally and 58 (13,5%) with clinical symptoms. Overall, family screening was more common in ERKReg than in ADPedKD (62% vs 48%) and prenatal diagnosis less common (see Table 2). However, the combined group of family screening and prenatal presentation was similar in both (64% vs 62%) and not all patients with prenatal abnormalities were assigned to prenatal diagnosis by the ERKReg investigators. The combined group of incidental finding and diagnosis due to symptoms (which can be hard to distinguish in clinical practice) were also similar in ERKReg and ADPedKD at 36% and 39%. For the RaDaR cohort, no details on mode of presentation were available.

The mean age at diagnosis in the ADPedKD cohort was 5.7 years (\pm 5.2 years, median 4.9, IQR 0.5 – 9.8, including prenatal diagnoses). Children identified by family screening had a lower age at diagnosis than those with symptomatic and incidental diagnosis (6.1 \pm 4.8 vs. 7.0 \pm 5.6 and 7.2 \pm 5.2 years, P = 0.01, see Table 3). A linear mixed model with country added as a random effect restricted to the cases with postnatal diagnosis also revealed a significantly higher age at diagnosis in children diagnosed by postnatal incidental findings compared to family screening (P < 0.01). Children diagnosed by family screening were thus usually below the age of consent, i.e. at diagnosis only 3/407 (0.7%) were aged above 16 years, 24/407 (5.9%) above 14 years and 46/407 (11.3%) above 12 years. A multivariable linear mixed model in the ADPedKD group showed no significant relation of positive family history, performance of a genetic test or healthcare spendings with age at diagnosis (see Supplementary Table S2).

Mean age at diagnosis was higher in both RaDaR and ERKReg cohorts at 10.5 and 9.3 years (see Table 2). This was not exclusively due to the lower proportion of prenatal diagnoses in ERKReg, because the mean age in the other mode of presentation-subgroups was also higher in ERKReg than in ADPedKD (see Table 3). In ERKReg 567 (69%) of patients were diagnosed before the age of 14 years; while 162/825 (20%) were over 16 years and 86/825 (8%) over 18 years old at diagnosis.

Among the 129 patients with prenatal diagnosis in ADPedKD, the most common anomalies were hyperechogenic kidneys (52/81 (64%)), kidney cysts (55/98 (56%)) and

enlarged kidneys (43/85 (51%)), but abnormal quantity of amniotic fluid was uncommon (13/81 (16%)). Of the 73 children who were diagnosed with signs or symptoms of ADPKD, the primary reason was usually urological: 16 (22%) with urinary tract infection, 16 (22%) with flank/back pain, 12 (16%) with hematuria, 5 (7%) with enuresis/dysfunctional voiding, 2 (2.7%) with urolithiasis and 1 (1.4%) with cyst complications. Kidney-related morbidities were less commonly the primary reason for presentation: 10 (14%) with hypertension and 2 (2.7%) with proteinuria. However, among the 376 children for whom the local investigator considered the diagnosis to have been an incidental finding, similar complaints (except cyst complications) were also common (see Supplementary Table S3).

Temporal changes

Prenatal diagnoses were very uncommon in children born before 2000 and increased until at least 2015 (Figure 1a and b). The proportion of diagnoses by family screening decreased slightly over time to be replaced mainly by prenatal diagnoses. Thus, year of birth was the most significant predictor of age at diagnosis in the multivariate model (see Supplementary Table S2). There was no association between year of diagnosis and the likelihood of genetic testing (P = 0.56). In comparison, in ERKReg more patients were born before 2000 but the temporal trends were similar with less family screening over time and a significant increase of prenatal diagnoses after 2010 (Figure 1c and d). Thus, for both registries most patients diagnosed prenatally were born after 2005 and are therefore currently still under 18 years old.

Geographical variations

The most pronounced regional differences between were the rates of genetic testing as shown in Figure 2 (by world region) and Supplementary Figure S1 (by country). Within ADPedKD there was a significant overall variability (P < 0.0001 for the likelihood-ratio test comparing the deviance of two nested models). Of the 12 countries represented both in ERKReg and ADPedKD the empirical Bayes estimates overlapped in all but 2 (Belgium and Italy), with higher rates in ADPedKD than ERKReg in both.

Both in ADPedKD and ERKReg cohorts, lower rates of genetic testing were observed in regions with lower health expenditure (in USD per capita, e.g. highest in Western Europe, followed by Southern Europe and Eastern Europe, see Figure 2 and compare mean healthcare expenditure in Table 1). This was also true on a country level (see Figure 3a and b). When correlation was tested with a generalized linear mixed model, this showed a significant association within the ERKReg cohort (odds ratio per 100USD/capita = 1.030 (95% confidence interval: 1.011 to 1.049), P = 0.002), and a very similar, but non-significant association in the ADPedKD cohort (odds ratio per 100USD/capita and year = 1.028 (95% confidence interval: 0.996 to 1.060), P = 0.09). As is visible in Figure 3a, this can be attributed firstly to the inclusion of Switzerland and the US, which both have extremely high health expenditure, but low levels of genetic testing (in the US it is not covered by health insurance and in Switzerland costs of genetic testing are only covered after individual application). As shown in supplemental Figure S2, the exclusion of the US and Switzerland resulted in a significant association of genetic testing with GDP in the mixed linear model ((odds ratio per 100USD/capita and year = 1.081 (95% confidence interval: 1.028 to 1.137), p = 0.0024).

There was little regional variation in the proportion of prenatal diagnoses (see Figure 4), but the postnatal modes of presentation differed significantly between countries, most so for the percentage of diagnoses by family screening (see Figure 5). However, family screening was less variable when countries were grouped by world region (see Figure S3). Both within the ADPedKD and within the ERKReg cohort, the country of origin had a significant effect in the general mixed linear model for the mode of presentation (both P < 0.0001). Regional variations in proportion of children diagnosed by incidental finding with symptoms are shown in Supplementary Figure S4.

The empirical Bayes estimates for mean age at diagnosis per world region are shown in Figure 6. There was some heterogeneity, but no clear geographical pattern. While there was a significant correlation in the uni- and multivariable models between age at diagnosis and country of residence (P < 0.001 and P = 0.0001), this explained only 1.03% and 1.49% of the overall variability in the uni- and multivariable model respectively (Supplementary Table S2).

DISCUSSION

We report on the initial diagnosis of a total of over 2100 children and young persons with ADPKD who come from 32 countries on four continents, covering 24 high-income and 8 middle-income countries with the aim of assessing regional and temporal influences on current modes of diagnosis.

The first finding was that a high proportion of children was diagnosed by family screening (48% in ADPedKD and 62% in ERKReg), with a slight tendency to decrease over time to the benefit of prenatal diagnosis in both registries. While there are multiple ethical, psychological and legal challenges around actively screening for a non-curable disease in children with a lively debate since the 1990s³³⁻³⁶, and the medical benefit is unclear³⁷, it appears that in real-life asymptomatic screening has nevertheless been done for some years, and about half the cases of pediatric ADPKD are diagnosed in this way. We postulate that it is a consequence of both the considerable stress some parents experience of not knowing about a child's transmission status³⁸ as well as their desire to be pro-active in contributing to the best outcome for their children. With the emerging data on the relevance of hypertension and/or proteinuria for the progression of kidney disease in ADPKD, a novel potential for early intervention is becoming more widely perceived. Indeed, ADPKD can no longer be regarded as untreatable, despite still being incurable. As emphasized by international consensus statements^{18,19} and the KDIGO 2025 Clinical Practice Guideline for the Evaluation, Management, and Treatment of ADPKD^{39(especially Chapter 7)}, it is important to discuss the pros and cons of pre-symptomatic screening of at-risk children and the potential implications of a positive diagnosis which are detailed in depth in the KDIGO guidelines (summarized in Table 3 of reference ³⁹⁾ with both parents and children in an age-appropriate manner. While health-care professionals should counsel families about the potential adverse effects of establishing a

diagnosis such as psychological burden and difficulties with private insurance or employment, they should also respectfully acknowledge families' wishes for diagnostic clarity which may affect further medical management, family planning, and provide opportunity and motivation for early lifestyle interventions. We emphasize the importance of informed and shared decision making by weighing the benefits and burdens of establishing a diagnosis while considering the child and family's values and situation. Screening for hypertension and proteinuria alone, without actively doing abdominal ultrasound, is a viable option but performing presymptomatic genetic testing or ultrasound should also be discussed¹⁷.

There was wide variation between countries in the proportion of screening diagnoses, that was not so evident when they were combined into regions of similar healthcare expenditure. We consider this to reflect the impact of local healthcare practices and ethical frameworks on physicians' and families' attitudes towards active screening in children of affected parents, which was also evident in a detailed patient focus group study from three different countries³⁸. There was no correlation of screening diagnoses to rates of genetic testing, probably because abdominal ultrasound is an alternative (if less reliable) method of diagnosis in childhood. Therefore, ethical and local healthcare frameworks are probably more important than economical constraints on genetic testing on determining rate of screening diagnoses.

The finding that 10-20% of cases were diagnosed prenatally may seem surprising for a disorder commonly considered as an "adult" disease, but this was remarkably similar between countries. Prenatal diagnoses became increasingly common since 2000 in both the ADPedKD and ERKReg cohorts, reflecting both the rapid development of prenatal ultrasound since the 1990s⁴⁰ and reports that fetuses with ADPKD primarily present with hyperechogenic and enlarged kidneys rather than classical cysts^{41–43}. A significant proportion of children in ERKReg had prenatal abnormalities recorded yet were not considered to be prenatally diagnosis by the local investigator. This may explain the slightly later rise of prenatal diagnoses in ERKReg. The high rate of prenatal diagnoses demonstrates that there is a high need of counselling of (prospective) parents affected by ADPKD, which is also stressed in the new KDIGO guidelines on ADPKD³⁹. From our data, we are unable to differentiate deliberate prenatal screening diagnoses from 'accidental findings' nor to estimate the number of terminations of pregnancies because of ADPKD. As high-resolution ultrasound is now part of routine pregnancy care in most of the countries represented here, families who consciously chose not to seek the child's (or fetus') transmission status, should be counselled to disclose this to the prenatal sonographer prior to examination, in the same way they should advise the sonographer at other postnatal abdominal ultrasound examinations (e.g. for abdominal trauma). In future analyses of the registry, we hope to characterize patients with prenatal abnormalities more fully in order to differentiate features that are risk-factors for very-early onset or rapidly progressive disease and those which are merely markers of disease transmission. As most patients with prenatal diagnoses were born after 2005, they are currently still under the age of 18 years and will be seen mainly in pediatric centers, which may explain the differing experiences of pediatric vs adult nephrologists when managing patients with "typical" ADPKD. However, this group of patients with prenatal diagnoses can be expected to reach clinics for adult nephrology within the next years.

The proportion of children who received genetic testing showed wide regional variation. This seems to be at least partially due to the cost of genetic testing as it was less common in countries with lower absolute healthcare expenditure (Figure 3). That genetic testing was not uniformly high in countries with high healthcare expenditure may reflect that it is not uniformly covered by insurance companies (e.g. in the US), varying cultural and religious beliefs and wide international discrepancy in legislation around genetic and genomic testing. For example, in some countries individuals do not have to disclose predictive genetic test results to private insurance companies, while in others the law prohibits molecular testing of minors affected with monogenic disease unless there is a benefit for themselves or related individuals in their family. Thus, the local healthcare systems may affect the decisions on screening taken by the families and clinicians in multiple ways. Interestingly, there was no increase in the proportion

of children receiving genetic testing over time despite the rapid advances in routine genetical testing in the last two decades, which may be due to the fact that *PKD1* and *PKD2* remain more challenging to sequence than other genes, that genotyping is still not common practice in adult nephrology (as the diagnosis is mainly based on ultrasound findings and family history¹²) and that accessibility is still limited in some countries. Knowledge about local rates of genetic testing may help national patient advocacy groups, medical organizations and/or planners in the healthcare to move towards improving patient care.

Children who were diagnosed with ADPKD due to signs or symptoms, presented with similar problems to those in previous smaller case series, which mainly reported symptoms during the course of the disease rather than exclusively at time of presentation^{37,41,44–47}. The increase in postnatal incidental diagnoses from 2001 to 2021 may reflect improvements in quality and availability of pediatric ultrasound, which some authors have suggested may increase the rate of kidney incidental findings⁴⁸. However, we would be hesitant to infer this, especially in children investigated for abdominal pain, as it can be very hard to distinguish a symptomatic diagnosis of ADPKD from an incidental finding and even a classical sign of ADPKD such as hypertension may have an alternate underlying cause. We respected the treating physician's classification of mode of presentation as either postnatal incidental finding or diagnosis with symptoms, but postulate that there is considerable overlap between both groups.

Limitations of this study are related to the inherent nature of registry data as being retrospective and with unknown bias as neither randomly sampled nor covering the entire population. Therefore, it is impossible to separate whether the rise in absolute patient numbers over time was due to the success of the registry or a real increase in children diagnosed with ADPKD due to increasing awareness of treatable disease manifestations during childhood. Also, we do not have data on terminations of pregnancy or pre-implantation genetic diagnosis of ADPKD and cannot comment on their effect on the ADPKD population. The fact that within each world region ADPedKD and ERKReg did not always cover identical countries limits the statistical comparability. This applies mainly to Europe-North, which in ADPedKD included only Lithuania as UK data were included in RaDaR, while for ERKReg it included Ireland, Estonia, Latvia and Lithuania. Other limitations were linked to clinical challenges in pediatric ADPKD, such as difficulty in reliably distinguishing whether an individual's symptoms (e.g. flank pain) are attributable to ADPKD or an incidental finding. Therefore, we draw no conclusions from the comparison of children diagnosed with ADPKD due to symptoms vs incidental findings, as it is probably not meaningful. We also did not perform a comparison of larger vs. smaller centers within countries, as this is mainly of local interest. Internationally, coverage of middle-income countries was much more limited than that of high-income countries, even though collaboration with centers in low- and middle-income countries was actively sought. Regrettably, no patients from low-income countries were included here, as at the time of data analysis only few centers from low-income countries had joined the ADPedKD and only started including patients later on. We hope to be able to fill the gap of data on whether and how ADPKD is diagnosed in lowincome countries in the future. Especially in resource-limited settings, provision of advanced nephrological care and participation in investigator-led research activities remains challenging. However, as the largest pediatric ADPKD cohort to date, this study provides an international reflection of ADPKD in children and demonstrates the success of multi-center cooperation. Future perspectives for the multi-national ADPedKD registry, will be to analyze prenatal presentations in more detail, examine for genotype-phenotype correlations and to develop predictive models for identifying children at-risk of fast progression and who are likely to benefit from active therapy 15,49 .

In summary, since the 1990s a high and stable proportion of children with ADPKD was diagnosed by active screening, indicating that a significant number of parents and health care professionals opt for this management choice. Regional variations in mode and age of presentation were more pronounced than temporal variations, thus technical advances seem to play a smaller role than the cultural attitudes within healthcare systems and the economic and legal frameworks in which they operate. Patients with a prenatal diagnosis of ADPKD are starting to constitute a significant proportion of pediatric cohorts and counseling families will be very important.

DISCLOSURE STATEMENT

All the authors declared no conflicts of interest.

SUPPLEMENTARY MATERIAL

Supplementary material is available online at <u>www.kidney-international.org</u>.

Supplementary Table S1. Number of patients included per country.

Supplementary Table S2. Results from a multivariable linear mixed models for age at diagnosis with country as a random effect (ADPedKD cohort).

Supplementary Table S3. Symptoms and signs present at diagnosis of ADPKD in the subgroups of children with symptomatic ADPKD and incidental finding.

Supplementary Figure S1. Empirical Bayes country-specific estimates for relative likelihood of genetic testing by country for the (a) ADPedKD and (b) ERKReg cohorts.

Supplementary Figure S2. Likelihood of genetic testing being performed, as predicted by a generalized linear mixed model (binary logistic regression) with country added as a random effect in the ADPedKD cohort (without USA and Switzerland).

Supplementary Figure S3. Empirical Bayes country-specific estimates for relative likelihood of diagnosis of ADPKD as postnatal incidental finding by world region in the (a) ADPedKD cohort and (b) ERKReg cohort.

Supplementary Figure S4. Empirical Bayes country-specific estimates for relative likelihood

of diagnosis of ADPKD as a result of postnatal incidental finding (a and b) or clinical symptoms

per country in the (c and d) per country in the (a and c) ADPedKD cohort and (b and d) ERKReg

cohort.

REFERENCES

- Spithoven EM, Kramer A, Meijer E, et al. Renal replacement therapy for autosomal dominant polycystic kidney disease (ADPKD) in Europe: prevalence and survival--an analysis of data from the ERA-EDTA Registry. *Nephrol Dial Transplant*. 2014;29 Suppl 4(Suppl 4):iv15-25. doi:10.1093/ndt/gfu017
- Cloutier M, Manceur AM, Guerin A, Aigbogun MS, Oberdhan D, Gauthier-Loiselle M. The societal economic burden of autosomal dominant polycystic kidney disease in the United States. *BMC Health Serv Res.* 2020;20(1):126. doi:10.1186/s12913-020-4974-4
- 3. Brunelli SM, Blanchette CM, Claxton AJ, Roy D, Rossetti S, Gutierrez B. Endstage renal disease in autosomal dominant polycystic kidney disease: a comparison of dialysis-related utilization and costs with other chronic kidney diseases. *Clinicoecon Outcomes Res.* 2015;7:65-72. doi:10.2147/CEOR.S76269
- 4. Lentine KL, Xiao H, Machnicki G, Gheorghian A, Schnitzler MA. Renal function and healthcare costs in patients with polycystic kidney disease. *Clin J Am Soc Nephrol*. 2010;5(8):1471-1479. doi:10.2215/CJN.00780110
- 5. Chauvet V, Qian F, Boute N, et al. Expression of PKD1 and PKD2 transcripts and proteins in human embryo and during normal kidney development. *Am J Pathol.* 2002;160(3):973-983. doi:10.1016/S0002-9440(10)64919-X
- Martínez V, Furlano M, Sans L, et al. Autosomal dominant polycystic kidney disease in young adults. *Clin Kidney J*. 2023;16(6):985-995. doi:10.1093/ckj/sfac251
- Halawi AA, Burgmaier K, Buescher AK, et al. Clinical Characteristics and Courses of Patients With Autosomal Recessive Polycystic Kidney Disease-Mimicking Phenocopies. *Kidney Int Rep.* 2023;8(7):1449-1454. doi:10.1016/j.ekir.2023.04.006
- 8. De Rechter S, Breysem L, Mekahli D. Is Autosomal Dominant Polycystic Kidney Disease Becoming a Pediatric Disorder? *Front Pediatr.* 2017;5:272. doi:10.3389/fped.2017.00272
- 9. Shamshirsaz A, Bekheirnia RM, Kamgar M, et al. Autosomal-dominant polycystic kidney disease in infancy and childhood: Progression and outcome. *Kidney International*. 2005;68(5):2218-2224. doi:10.1111/j.1523-1755.2005.00678.x

- 10. Nowak KL, Cadnapaphornchai MA, Chonchol MB, Schrier RW, Gitomer B. Long-Term Outcomes in Patients with Very-Early Onset Autosomal Dominant Polycystic Kidney Disease. *AJN*. 2016;44(3):171-178. doi:10.1159/000448695
- 11. Bergmann C, Zerres K. Early manifestations of polycystic kidney disease. *Lancet*. 2007;369(9580):2157. doi:10.1016/S0140-6736(07)61005-8
- Cornec-Le Gall E, Torres VE, Harris PC. Genetic Complexity of Autosomal Dominant Polycystic Kidney and Liver Diseases. J Am Soc Nephrol. 2018;29(1):13-23. doi:10.1681/ASN.2017050483
- 13. Lanktree MB, Guiard E, Li W, et al. Intrafamilial Variability of ADPKD. *Kidney Int Rep.* 2019;4(7):995-1003. doi:10.1016/j.ekir.2019.04.018
- Liebau MC, Mekahli D, Perrone R, Soyfer B, Fedeles S. Polycystic Kidney Disease Drug Development: A Conference Report. *Kidney Med*. 2023;5(3):100596. doi:10.1016/j.xkme.2022.100596
- Mekahli D, Womack H, Dahl NK. Perspectives on Drug Development in Early ADPKD. *Clin J Am Soc Nephrol*. 2022;17(10):1555-1558. doi:10.2215/CJN.05190422
- Breysem L, De Keyzer F, Schellekens P, et al. Risk Severity Model for Pediatric Autosomal Dominant Polycystic Kidney Disease Using 3D Ultrasound Volumetry. *Clin J Am Soc Nephrol.* 2023;18(5):581-591. doi:10.2215/CJN.00000000000122
- 17. Gimpel C, Bergmann C, Mekahli D. The wind of change in the management of autosomal dominant polycystic kidney disease in childhood. *Pediatr Nephrol*. Published online March 7, 2021. doi:10.1007/s00467-021-04974-4
- Gimpel C, Bergmann C, Bockenhauer D, et al. International consensus statement on the diagnosis and management of autosomal dominant polycystic kidney disease in children and young people. *Nat Rev Nephrol.* 2019;15(11):713-726. doi:10.1038/s41581-019-0155-2
- Chapman AB, Devuyst O, Eckardt KU, et al. Autosomal-dominant polycystic kidney disease (ADPKD): executive summary from a Kidney Disease: Improving Global Outcomes (KDIGO) Controversies Conference. *Kidney Int.* 2015;88(1):17-27. doi:10.1038/ki.2015.59
- Harris T, Sandford R, de Coninck B, et al. European ADPKD Forum multidisciplinary position statement on autosomal dominant polycystic kidney disease care. European ADPKD Forum and Multispecialist Roundtable participants. *Nephrol Dial Transplant*. 2018;33(4):563-573. doi:10.1093/ndt/gfx327
- 21. Rangan GK, Savige J. Introduction to the KHA-CARI Guidelines on ADPKD. Semin Nephrol. 2015;35(6):521-523. doi:10.1016/j.semnephrol.2015.10.001
- 22. Wong K, Pitcher D, Braddon F, et al. The National Registry of Rare Kidney Diseases (RaDaR): design, recruitment, and cross-sectional analyses of 25,880

adults and children with rare kidney diseases in the UK. Published online September 25, 2023:2023.09.24.23296009. doi:10.1101/2023.09.24.23296009

- 23. Bassanese G, Wlodkowski T, Servais A, et al. The European Rare Kidney Disease Registry (ERKReg): objectives, design and initial results. *Orphanet J Rare Dis.* 2021;16(1):251. doi:10.1186/s13023-021-01872-8
- 24. De Rechter S, Bockenhauer D, Guay-Woodford LM, et al. ADPedKD: A Global Online Platform on the Management of Children With ADPKD. *Kidney Int Rep.* 2019;4(9):1271-1284. doi:10.1016/j.ekir.2019.05.015
- 25. Wong K, Pitcher D, Braddon F, et al. Effects of rare kidney diseases on kidney failure: a longitudinal analysis of the UK National Registry of Rare Kidney Diseases (RaDaR) cohort. *Lancet*. 2024;403(10433):1279-1289. doi:10.1016/S0140-6736(23)02843-X
- 26. Alzarka B, Morizono H, Bollman JW, Kim D, Guay-Woodford LM. Design and Implementation of the Hepatorenal Fibrocystic Disease Core Center Clinical Database: A Centralized Resource for Characterizing Autosomal Recessive Polycystic Kidney Disease and Other Hepatorenal Fibrocystic Diseases. *Frontiers in Pediatrics*. 2017;5. Accessed August 2, 2023. https://www.frontiersin.org/articles/10.3389/fped.2017.00080
- 27. Jayasinghe K, Stark Z, Patel C, et al. Comprehensive evaluation of a prospective Australian patient cohort with suspected genetic kidney disease undergoing clinical genomic testing: a study protocol. *BMJ Open*. 2019;9(8):e029541. doi:10.1136/bmjopen-2019-029541
- 28. WHO regions. WHO. Accessed December 13, 2024. https://apps.who.int/gho/data/node.metadata.REGION?lang=en
- 29. Standard Country or Area Codes for Statistical Use (M49 standard). Accessed December 13, 2024. https://unstats.un.org/unsd/methodology/m49/
- 30. Vardell E. Global Health Observatory Data Repository. *Med Ref Serv Q.* 2020;39(1):67-74. doi:10.1080/02763869.2019.1693231
- 31. World Health Organisation. Current health expenditure as percentage of gross domestic product (GDP). Global Health Observatory data repository 2019. https://apps.who.int/gho/data/view.main.GHEDCHEGDPSHA2011v Accessed online 11th August 2022
- 32. Verbeke G, Molenberghs G. *Linear Mixed Models for Longitudinal Data*. Springer; 2000.
- Sujansky E, Kreutzer SB, Johnson AM, Lezotte DC, Schrier RW, Gabow PA. Attitudes of at-risk and affected individuals regarding presymptomatic testing for autosomal dominant polycystic kidney disease. *Am J Med Genet*. 1990;35(4):510-515. doi:10.1002/ajmg.1320350413

- 34. Zidovská J, Merta M, Kapras J. [The attitude of family members with polycystic kidney disease to the disease and presymptomatic testing]. *Cas Lek Cesk*. 1994;133(6):184-187.
- 35. De Rechter S, Kringen J, Janssens P, et al. Clinicians' attitude towards family planning and timing of diagnosis in autosomal dominant polycystic kidney disease. *PLoS ONE*. 2017;12(9):e0185779. doi:10.1371/journal.pone.0185779
- 36. Harris T. Is It Ethical to Test Apparently "Healthy" Children for Autosomal Dominant Polycystic Kidney Disease and Risk Medicalizing Thousands? *Front Pediatr.* 2018;5:291. doi:10.3389/fped.2017.00291
- 37. Mekahli D, Woolf AS, Bockenhauer D. Similar renal outcomes in children with ADPKD diagnosed by screening or presenting with symptoms. *Pediatr Nephrol.* 2010;25(11):2275-2282. doi:10.1007/s00467-010-1617-8
- Logeman C, Cho Y, Sautenet B, et al. "A sword of Damocles": patient and caregiver beliefs, attitudes and perspectives on presymptomatic testing for autosomal dominant polycystic kidney disease: a focus group study. *BMJ Open*. 2020;10(10):e038005. doi:10.1136/bmjopen-2020-038005
- 39. Devuyst O, Torres VE. 2025 Clinical Practice Guideline for the Evaluation, Management, and Treatment of Autosomal Dominant Polycystic Kidney Disease (ADPKD). *Kidney International*. 2025;in press.
- 40. Morris JK, Springett AL, Greenlees R, et al. Trends in congenital anomalies in Europe from 1980 to 2012. *PLoS One*. 2018;13(4):e0194986. doi:10.1371/journal.pone.0194986
- 41. Fick GM, Duley IT, Johnson AM, Strain JD, Manco-Johnson ML, Gabow PA. The spectrum of autosomal dominant polycystic kidney disease in children. *JASN*. 1994;4(9):1654-1660.
- 42. Brun M, Maugey-Laulom B, Eurin D, Didier F, Avni EF. Prenatal sonographic patterns in autosomal dominant polycystic kidney disease: a multicenter study. *Ultrasound Obstet Gynecol.* 2004;24(1):55-61. doi:10.1002/uog.1098
- 43. Boyer O, Gagnadoux MF, Guest G, et al. Prognosis of autosomal dominant polycystic kidney disease diagnosed in utero or at birth. *Pediatr Nephrol.* 2007;22(3):380-388. doi:10.1007/s00467-006-0327-8
- 44. Marlais M, Cuthell O, Langan D, Dudley J, Sinha MD, Winyard PJD. Hypertension in autosomal dominant polycystic kidney disease: a meta-analysis. *Archives of Disease in Childhood*. 2016;101(12):1142-1147. doi:10.1136/archdischild-2015-310221
- 45. Massella L, Mekahli D, Paripović D, et al. Prevalence of Hypertension in Children with Early-Stage ADPKD. *Clin J Am Soc Nephrol.* 2018;13(6):874-883. doi:10.2215/CJN.11401017

- 46. Fencl F, Janda J, Bláhová K, et al. Genotype–phenotype correlation in children with autosomal dominant polycystic kidney disease. *Pediatr Nephrol.* 2009;24(5):983-989. doi:10.1007/s00467-008-1090-9
- 47. Zittema D, Boertien WE, Beek AP van, et al. Vasopressin, Copeptin, and Renal Concentrating Capacity in Patients with Autosomal Dominant Polycystic Kidney Disease without Renal Impairment. *CJASN*. 2012;7(6):906-913. doi:10.2215/CJN.11311111
- 48. O'Kelly F, McAlpine K, Abdeen N, Keays MA, Leonard MP, Guerra LA. The Prevalence, Clinicodemographics, and Outcomes of Incidental and Symptomatic Renal Cysts in a Pediatric Cohort Undergoing Ultrasonography. *J Urol.* 2019;202(2):394-399. doi:10.1097/JU.00000000000264
- Ostroff C, Perrone RD, Czerwiec FS. Current Challenges and Perspectives on Developing a Clinical Trial Design for ADPKD. *Clin J Am Soc Nephrol.* 2022;17(10):1559-1562. doi:10.2215/CJN.05360522

ACKNOWLEDGEMENTS

The authors would like to thank the patients enrolled in the ADPedKD, RaDaR and ERKReg Registry and their families as well as all the participating centers for their active collaboration in this project.

The authors acknowledge the ADPedKD study group: B. Chiodini, J. Cruz Len Aguilera, F. Durao, L. Eid, M. Espino-Hernández, M. Furlana, M.S. Geysen, M. Giordano, N. Godefroid, J. Groothoff, P. Hansen, N. Hooman, B. Josselin, M. Kemper, I. Liu, J. Lombet, N. Segers, R. Sinha, N. A. Soliman, S. Stabouli, R. Stroescu, T. Sulakova, A. Szmigielska, K. Taranta-Janusz, A0 Teixeira, M. Tkaczyk, A. Zvenigorodska.

The authors acknowledge the RaDaR ADPKD Rare Disease Clinical Study Group: E. Asgari, C. Bingham, K. Bramham, J. Fotheringham, M. Gittus, T. Harris, K. Hillman, N. Inston, L. Kerecuk, ACM. Ong, R. Sandford, J. A. Sayer, R. Simms, M. Sinha, S. Srivastava, R.Steenkamp, D. C. Wheeler, P. Wilson, P. Winyard, G. Wood

The authors acknowledge the ERKReg collaborators: A. Adoberg, A. C. Afonso, I. Andersone, K. Arbeiter, A. Awan, B. Bammens, A. Bouts, A. Buescher, J. Calado, L. R. Claus, A. Debska-Slizien, J. de Fallois, M. Furlano, M. Gabriele, D. Haffner, L. Koster-Kamphuis, F. Lugani, M. Miglinas, J. Oh, M. Pawlak-Bratkowska, D. Roccatello, E. E. Rusu, M. Santo Stefano, F. Scolari, E. Siomou, A. Skoberne, A. Debska-Slizien, A. Szabó, E. Vidal, M. Weitz

The ADPedKD initiative has been supported by the European Renal Association (ERA), the European Society of Pediatric Nephrology (ESPN) and its Working Group on Inherited Kidney Diseases, the European Rare Kidney Disease Reference Network (ERKNet) and the Fund of UZ Leuven. ERKNet is funded by the European Union within the framework of the EU4Health Program (101085068).

DM is supported by the Research Foundation Flanders (FWO) (G0C8920N), the clinical senior research grant for DM (1804123N) and the University Hospitals Leuven. LGW is supported by the UAB Hepatorenal Fibrocystic Disease Core Center (National Institutes of Health (NIH)/National Institute of Diabetes and Digestive and Kidney Diseases P30 DK074038), the Polycystic Kidney Disease Foundation, and the Clinical and Translational Science Institute at Children's National (CTSI-CN; NIH/National Center for Advancing Translational Sciences UL1TR001876). MCL is supported by the Medical Faculty of the University of Cologne and the Neocyst Consortium funded by the German Ministry for Education and Research (BMBF, 01GM2203B). CG and FS are supported by the Neocyst Consortium funded by the German Ministry for Education and Research (BMBF, 01GM2203D).

DATA AVAILABILITY STATEMENT

Data of the ADPedKD, ERKReg and RaDaR registries is not publicly available. Access can be requested from the respective coordinators (see author list).

World	Total number of	Total	Bange of national	Mean weighted	
region	countries (high/	number of	nange of national	healthcare expenditure**	
	middle income)	patients	healthcare expenditures *		
SEAR	1 (0/1)	5	58	58	
EUR-East	7 (4/3)	262	157-1,284	834	
EMR	3 (1/2)	13	174-1,439	916	
EUR-South	7 (5/2)	678	447-2,676	2,115	
EUR-North	5 (5/0)	401	776-4,561	4,112	
EUR-West	6 (6/0)	759	4,199-9,383	4,476	
WPR	2 (2/0)	20	2,279-5,324	4,868	
AMR	1 (1/0)	16	9,244	9,244	

Table 1 | Number of countries, patients and healthcare expenditure by world region

* healthcare expenditure in USD per capita in 2015³¹

** weighted for number of patients from each country

WHO regions (and countries represented in the present cohorts):
SEAR: South East Asian Region (India)
EUR: European Region (Europe-East: Belarus, Czech Republic, Hungary, Poland, Romania, Russia, Ukraine. Europe-South: Greece, Italy, Portugal, Serbia, Slovenia, Spain, Turkey. Europe-North: Estonia, Ireland, Latvia, Lithuania, UK. Europe-West: Austria, Belgium, France, Germany, Netherlands, Switzerland)
EMR: Eastern Mediterranean Region (Egypt, Iran, UAE)
WPR: Western Pacific Region (Australia, Singapore)
AMR: Region of the Americas (USA)

	ADPedKD	RaDaR	ERKReg
Pediatric ADPKD population size, N	1060	269	825
Genetic test performed			
Yes, N	547 (52%)	117 (43%)	335 (41%)
No, N	454 (43%)	Not recorded	Not recorded
Unknown, N	59 (6%)	152 (57%)	490 (59%)
Genetic results (information available), N	547	117 ¹	335 ¹
No information or NMD in ADPKD genes, N	22 (4%)	36 (31%)	61 (18%)
PKD1 causative variant, N	470 (86%)	73 (62%)	245 (73%)
PKD2 causative variant, N	45 (8%)	6 (5%)	27 (8%)
PKD1+TSC2 causative variant, N	6 (1%)	2 (2%)	0 (0%)
Other causative variant, N	4 (1%)	0 (0.0)	2 (1%)
Ratio of PKD1 to PKD2 variants	10.6 : 1	12.5 : 1	9:1
Diagnosis (information available), N	950	139	596
Prenatal, N	129 (14%)	15 (11%)	13 (2%)
Family screening, N	455 (48%)	unknown	370 (62%)
Incidental finding, N	293 (31%)	unknown	126 (21%)
With symptoms, N	73 (8%)	unknown	87 (15%)
Age at diagnosis (information available), N	945	269	825
Mean age in years (SD)	5.7 (5.2)	10.5 (6.7)	9.3 (6.2)
Median age in years (IQR)	4.9 (0.5-9.8)	12.2 (4.4-16.5)	9.8 (3.5-15.0)
Min; max	-0.7 - 18.5	-0.7 - 19.0	-0.3 - 19.0

Table 2 | Demographical characteristics of the ADPedKD, RaDaR and ERKReg cohorts

ADPKD, autosomal dominant polycystic kidney disease; *IQR*, interquartile range; *Max*, maximum *Min*, minimum; *NMD*, no causative variant detected; *N*, number of observations with data; *SD*, standard deviation.

¹In the RaDaR and ERKReg cohorts, no distinction was available between no genetic test performed and unknown. Hence, all patients without genetic results were considered as no genetic test performed.

	ADPedKD			ERKReg		
	Ν	Mean	SD	Ν	Mean	SD
Total	945	5.7	5.2	825	9.3	6.2
Prenatal	129	-0.1	0.1	13	0.1	0.2
Family screening	455	6.1	4.8	370	10.4	6.1
Incidental finding	293	7.2	5.2	126	11.0	4.0
Clinical symptoms	73	7.0	5.6	87	8.8	5.9

Table 3 | Age at diagnosis (in years) in the ADPedKD and ERKReg cohorts as a function

 of mode of diagnosis

N, Number of observations; SD, standard deviation

Figure legends

Figure 1 | Distribution of mode of presentation of ADPKD per year of birth for (a) & (b)

ADPedKD cohort (*n*=945) and (c) & (d) ERKReg cohort (*n*=825)

Figure 2 | Bayes estimates for percentage of children receiving genetic testing by world

region. (a) ADPedKD cohort (b) ERKReg cohort.

The dashed line represents the mean over all regions. Dots and numbers indicate empirical Bayes

estimates for the probability of genetic testing in that region with horizontal lines and numbers in

brackets indicating the 95% confidence interval (CI).

<u>ADPedKD</u>

AMR: Region of the Americas (1 country: USA n=16) EMR: Eastern Mediterranean Region (3 countries: Egypt n=1; Iran n=4; United Arab Emirates n=7) EUR-East: European Region - East (6 countries: Belarus n = 11; Czech Republic n = 68; Poland n = 96; Romania n=13; Russia n=22; Ukraine n=4) EUR-North: European Region - North (1 country: Lithuania n=22) EUR-South: European Region - South (6 countries: Greece n= 5; Italy n= 172; Portugal n= 10; Serbia n= 40; Spain *n*= 25; Turkey *n*= 91) EUR-West: European Region - West (5 countries: Belgium n=155; France n=181; Germany n=18; Netherlands n=2; Switzerland n=14) SEAR: South East Asian Region (1 country: India n=5) WPR: Western Pacific Region (2 countries: Australia n=17; Singapore n=2) ERKReg EUR-East: European Region - East (4 countries: Hungary n= 1; Poland n= 26; Romania n= 14; Russia n= 4) EUR-North: European Region - North (4 countries: Estonia n= 1; Ireland n= 91; Latvia n= 3; Lithuania n= 11) EUR-South: European Region - South (5 countries: Greece n=1; Italy n=225; Portugal n=4; Slovenia n=138: Spain *n*= 36) EUR-West: European Region - West (5 countries: Austria n=13; Belgium n=68; France n=159; Germany n= 86: Netherlands n= 44)

Figure 3 | Likelihood of genetic testing being performed in children with ADPKD, as

predicted by a generalised linear mixed model (binary logistic regression) with country

added as a random effect in the (a) ADPedKD cohort and (b) ERKReg cohort.

Current health expenditure is expressed as USD per capita per year (as of 2015). Shaded area:

95% confidence interval.

For number of patients per country in both cohorts: see Supplementary Table S1.

Figure 4 | Bayes estimates for the relative likelihood of a prenatal diagnosis of ADPKD by

world region in the (a) ADPedKD cohort and (b) ERKReg cohort

The dashed line represents the mean over all regions. Dots and numbers indicate empirical Bayes

country-specific estimates for the probability of genetic testing with horizontal lines and numbers

in brackets indicating the 95% confidence interval (CI).

ADPedKD

EMR: Eastern Mediterranean Region (3 countries: Egypt n=1; Iran n=4; United Arab Emirates n=7) EUR-East: European Region - East (6 countries: Belarus n = 11; Czech Republic n = 68; Poland n = 89; Romania n=13; Russia n=12; Ukraine n=4) EUR-North: European Region - North (1 country: Lithuania n=20) EUR-South: European Region - South (6 countries: Greece n=5; Italy n=195; Portugal n=11; Serbia n=11; 39; Spain *n*= 25; Turkey *n*= 92) EUR-West: European Region - West (5 countries: Belgium n= 154; France n= 159; Germany n= 16; Netherlands n=2; Switzerland n=16) SEAR: South East Asian Region (1 country: India n=4) WPR: Western Pacific Region (1 country: Singapore n=3) ERKRea EUR-East: European Region - East (3 countries: Hungary n=1: Poland n=7: Romania n=10) EUR-North: European Region - North (4 countries: Estonia n=1; Ireland n=90; Latvia n=2; Lithuania n=110)EUR-South: European Region - South (4 countries: Italy n=141; Portugal n=1; Slovenia n=22; Spain n=122; Spain n=12227) EUR-West: European Region - West (5 countries: Austria n=5; Belgium n=59; France n=129; Germany n=53; Netherlands n=38)

Figure 6 | Bayes estimates for mean age (in years) at diagnosis of ADPKD in the (a)

ADPedKD cohort and (b) ERKReg cohort by world region

The dashed line represents the mean over all regions. Dots and numbers indicate empirical Bayes

estimates for the age at diagnosis in this world region with horizontal lines and numbers in

brackets indicating the 95% confidence interval (CI).

ADPedKD

AMR: Region of the Americas (1 country: USA n=15)

EMR: Eastern Mediterranean Region (3 countries: Egypt n=2; Iran n=3; United Arab Emirates n=7) EUR-East: European Region - East (6 countries: Belarus n=11; Czech Republic n=64; Poland n=82; Romania n=13; Russia n=20; Ukraine n=4) EUR North: European Region - North (1 country: Lithuania n=25)

EUR-North: European Region - North (1 country: Lithuania *n*= 25)

EUR-South: European Region - South (6 countries: Greece n=5; Italy n=179; Portugal n=11; Serbia n=39; Spain n=25; Turkey n=83)

EUR-West: European Region - West (5 countries: Belgium n= 144; France n= 164; Germany n= 8; Netherlands n= 2; Switzerland n= 15)

SEAR: South East Asian Region (1 country: India n=4)

WPR: Western Pacific Region (2 countries: Australia n=17; Singapore n=3) ERKReg:

EUR-East: European Region - East (4 countries: Hungary n=1; Poland n=26; Romania n=14; Russia n=4)

EUR-North: European Region - North (4 countries: Estonia n=1; Ireland n=91; Latvia n=3; Lithuania n=11)

EUR-South: European Region - South (5 countries: Greece n=1; Italy n=225; Portugal n=4; Slovenia n=38; Spain n=36)

EUR-West: European Region - West (5 countries: Austria n=13; Belgium n=68; France n=159; Germany n=86; Netherlands n=44)

A multi-national perspective on the presentation of childhood autosomal dominant polycystic kidney disease in high- and middle-income countries: insights from the ADPedKD, ERKReg and RaDaR registries

Supplementary Material

Supplementary Table S1. Number of patients included per country
Supplementary Table S2. Results from a multivariable linear mixed models for age at
diagnosis with country as a random effect (ADPedKD cohort)
Supplementary Table S3. Symptoms and signs present at diagnosis of ADPKD in the
subgroups of children with symptomatic ADPKD and incidental finding
Supplementary Figure S1. Empirical Bayes country-specific estimates for relative
likelihood of genetic testing being performed in children with ADPKD by country for
the (a) ADPedKD and (b) ERKReg cohorts 41
Supplementary Figure S2. Likelihood of genetic testing being performed, as predicted
by a generalized linear mixed model (binary logistic regression) with country
added as a random effect in the ADPedKD cohort (without USA and Switzerland)
Supplementary Figure S3. Empirical Bayes estimates for relative likelihood of
diagnosis of ADPKD as postnatal incidental finding by world region in the (a)
ADPedKD cohort and (b) ERKReg cohort 43
Supplementary Figure S4. Empiricial Bayes country-specific estimates for relative
likelihood of diagnosis of ADPKD as a result of postnatal incidental finding (a and
b) or clinical symptoms (c and d) per country in the (a and c) ADPedKD cohort and
(b and d) ERKReg cohort 44

Country	World region		PaDaP	EDKDog	WHO classification
Country	wond region	ADFEUND	naDan	Ennney	income group
Australia	WPR	17			High
Austria	EUR-West			13	High
Belarus	EUR-East	11			Upper middle
Belgium	EUR-West	156		68	High
Czech Republic	EUR-East	69			High
Egypt	EMR	2			Lower middle
Estonia	EUR-North			1	High
France	EUR-West	190		159	High
Germany	EUR-West	25		86	High
Greece	EUR-South	5		1	High
Hungary	EUR-East			1	High
India	SEAR	5			Lower middle
Iran	EMR	4			Lower middle
Ireland	EUR-North			91	High
Italy	EUR-South	195		225	High
Latvia	EUR-North			3	High
Lithuania	EUR-North	26		11	High
Netherlands	EUR-West	2		44	High
Poland	EUR-East	98		26	High
Portugal	EUR-South	11		4	High
Romania	EUR-East	13		14	High
Russia	EUR-East	22		4	Upper middle
Serbia	EUR-South	42			Upper middle
Singapore	WPR	3			High
Slovenia	EUR-South			38	High
Spain	EUR-South	25		36	High
Switzerland	EUR-West	16			High
Turkey	EUR-South	96			Upper middle
UAE	EMR	7			High
UK	EUR-North		269		High
Ukraine	EUR-East	4			Lower middle
USA	AMR	16			High

Supplementary Table S1. Number of patients included per country.

SEAR: South East Asian Region, EUR: European Region, EMR: Eastern Mediterranean Region, WPR: Western Pacific Region, AMR: Region of the Americas

UAE, United Arab Emirates; *UK*, United Kingdom; *USA*, United States of America. A multi-national perspective on the presentation of childhood autosomal dominant polycystic kidney disease in high- and middle-income countries: insights from the ADPedKD, ERKReg and RaDaR registries. Supplementary Material

		Estimate (SE)	P-value
Year of birth			<.0001
	<1991	11.593 (SE=0.947)	<.0001
	1991-1995	8.409 (SE=0.958)	<.0001
	1996-2000	8.483 (SE=0.679)	<.0001
	2001-2005	7.711 (SE=0.524)	<.0001
	2006-2010	4.395 (SE=0.518)	<.0001
	2011-2015	1.625 (SE=0.540)	0.0027
	>2015	#	
Genetic test performed			0.8743
	No	0.117 (SE=0.679)	0.8635
	Yes	-0.423 (SE=1.239)	0.7328
	Unknown	#	
Positive family history			0.3336
	No	0.351 (SE=0.363)	
	Yes	#	
Genetic grouping			0.0306
	No evidence or information on ADPDK related mutations	-7.759 (SE=4.320)	0.0729
	PKD1 or (PKD1+PDK2)	-7.660 (SE=4.196)	0.0682
	Only PDK2	-5.997 (SE=4.249)	0.1585
	PKD1+TSC2	-9.918 (SE=4.531)	0.0289
	Other	#	
Current Health Expenditure (per 100 dollar/person)		-0.014 (SE=0.015)	0.3693

Supplementary Table S2. Results from a multivariable linear mixed models for age at diagnosis with country as a random effect (ADPedKD cohort).

The multivariable linear mixed model was fitted on 913 observations. Children with prenatal diagnosis are excluded, as they have a different age at diagnosis per se.

Random effect of country: X^2 = 13.71, p-value=0.0001. Explained variability by random effect derived from the X^2 statistic: 1.49%.

ADPKD, autosomal dominant polycystic kidney disease; SE, standard error; #, reference category

Supplementary Table S3. Symptoms and signs present at diagnosis of ADPKD in the subgroups of children with symptomatic ADPKD and incidental finding.

	ADPKD diagnosed	ADPKD diagnosed as
	with symptoms / signs	an incidental finding
	N=73	N=376
Urinary tract infection	16 (22%)	41 (11%)
Enuresis or dysfunctional voiding	5 (7%)	7 (2%)
Flank/ back/ abdominal pain ¹	16 (22%)	84 (22%)
Urolithiasis	2 (3%)	4 (1%)
Hematuria	12 (16%)	13 (3%)
Cyst complications	1 (1%)	0
Hypertension	10 (14%)	6 (2%)
Proteinuria	2 (3%)	3 (1%)

Assignment to either diagnosis group (with symptoms/ incidental) as classified by local investigator.

¹Children who presented with abdominal pain for a named other reason (e.g. appendicitis, constipation, trauma, pancreatitis or malignancy are not included).

Supplementary Figure S1. Empirical Bayes country-specific estimates for relative likelihood of genetic testing being performed in children with ADPKD by country for the (a) ADPedKD and (b) ERKReg cohorts.



Dots/numbers indicate country-specific estimates for the probability of this mode of presentation with horizontal lines/numbers in brackets referring to the 95% confidence interval. The dashed line represents the mean over all countries

Supplementary Figure S2. Likelihood of genetic testing being performed, as predicted by a generalized linear mixed model (binary logistic regression) with country added as a random effect in the ADPedKD cohort (without USA and Switzerland)



Supplementary Figure S3. Empirical Bayes estimates for relative likelihood of diagnosis of ADPKD as postnatal incidental finding by world region in the (a) ADPedKD cohort and (b) ERKReg cohort



Dots/numbers indicate region-specific estimates for the probability of this mode of presentation with horizontal lines/numbers in brackets referring to the 95% confidence interval. The dashed line represents the mean over all regions. *CI*, confidence interval

Supplementary Figure S4. Empiricial Bayes country-specific estimates for relative likelihood of diagnosis of ADPKD as a result of postnatal incidental finding (a and b) or clinical symptoms (c and d) per country in the (a and c) ADPedKD cohort and (b and d) ERKReg cohort



Dots/numbers indicate country-specific estimates for the probability of this mode of presentation with horizontal lines/numbers in brackets referring to the 95% confidence interval. The dashed line represents the mean over all countries.

CI, confidence interval; *UAE*, United Arab Emirates; *UK*, United Kingdom; *USA*, United States of America.